Author’s response to reviews

Title: Coreopsis tinctoria Nutt ameliorates high glucose-induced renal fibrosis and inflammation via the TGF-β1/SMADS/AMPK/NF-κB pathways

Authors:

Lan Yao (ylyl1026@sina.com)
Jie Li (alexli91@yahoo.com)
Lin lin Li (442338720@qq.com)
Xin xia Li (lxx6668@163.com)
Rui Zhang (603210439@qq.com)
Yu jie Zhang (835580053@qq.com)
Xin Min Mao (mxm3277@139.com)

Version: 5 Date: 18 Oct 2018

Author’s response to reviews:

In the Authors' contributions section, please refer to all authors by their initials rather than full names. For example, Lan Y should be listed as LY. Yes all authors' name have been listed as LY (initials).

2. In your funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript. Please note this statement refers to disclosure of any competing interest. If the funding body did not play a role in these aspect of the study please include a statement indicating that the funder played no role in ‘the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript’.

The role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript has been stated carefully.

3. Please ensure that Figure 8 is referred to in the main body of the manuscript (for example in the Results or Discussion section). All figures and tables should be referred to in the main manuscript text. Yes, Figure 8 is referred to in the main body of Results.